Author’s response to reviews

Title: Molecular Characterization and Epidemiology of Carbapenem Non-susceptible Enterobacteriaceae Isolated from the Eastern Region of Heilongjiang Province, China

Authors:

XUE GONG (291135155@qq.com)
Jisheng Zhang (zhangjisheng215@126.com)
Shanshan Su (827691363@qq.com)
Yanjun Fu (fuyanjun1981@163.com)
Mingjia Bao (cmubmj@163.com)
Yong Wang (15331878255@163.com)
Xiaoli Zhang (jmszxl123@163.com)

Version: 5 Date: 27 Jul 2018

Author’s response to reviews:

Dear Edward Spofford,

Thank you for your suggestions on my manuscript. The following is my reply to the Ethics approval and consent to participate section. It has been modified in manuscript.

The study protocol was approved by the Ethics Committee of Jiamusi University Clinical Medical College for research. The committee’s reference number is 0326. Individual informed consent was waived by the ethics committee listed above because this study used currently existing sample collected during the course of routine medical care and did not pose any additional risks to the patients.

Best wishes,

Xiaoli Zhang